Author's response to reviews

Title: Bias in trials comparing paired continuous tests can cause researchers to choose the wrong screening modality

Authors:

Deborah H Glueck (Deborah.Glueck@ucdenver.edu)
Molly M Lamb (hgl9@cdc.gov)
Colin I Odonnell (Colin.Odonnell@ucdenver.edu)
Brandy M Ringham (Brandy.Ringham@ucdenver.edu)
John T Brinton (John.Brinton@ucdenver.edu)
Keith E Muller (Keith.Muller@biostat.ufl.edu)
John M Lewin (jlewin@divrad.com)
Todd A Alonzo (talonzo@childrensoncologygroup.org)
Etta D Pisano (etpisano@med.unc.edu)

Version: 4 Date: 9 January 2009

Author's response to reviews: see over
January 9, 2009
Re: MS # 3318613332079587

Dear Dr. Norton:

My co-authors and I were very pleased that our manuscript will be published in BMC Medical Research Methodology. I have made the formatting changes that you requested, as follows.

1. I have removed the authors’ qualifications from the manuscript.
2. I have re-ordered the figure citations, and the figures.
3. I have now placed the appendix before the acknowledgements section.

If you need any further changes, let me know.

Sincerely,

Deborah Glueck, Ph.D.

Assistant Professor
Department of Biostatistics
Colorado School of Public Health
University of Colorado Denver
Historic Building 500
Room W3125
13001 E. 17th Place
Mail Stop C245
Aurora, CO 80045